

April 2, 2014

Dear Editor,

Please find enclosed the edited manuscript in Word format (file name: 8158-review.doc).

**Title:** Social Determinants of Type 2 Diabetes and Health in the United States

**Author:** Myra L. Clark, Sharon W. Utz.

**Name of Journal:** *World Journal of Diabetes*

**ESPS Manuscript NO:** 8158

The manuscript has been improved according to the suggestions of reviewers:

1 Revision has been made according to the suggestions of the reviewers

(1) The manuscript title was changed to Social Determinants of type 2 Diabetes and Health in the United States as recommended by reviewer 2

(2) The exact reference for each study was marked in the results section as recommended by reviewer 2.

(3) A brief commentary about neonatal age and insulin resistance was added to the literature background as recommended by reviewer 1.

(4) A brief commentary on functional foods as nutraceuticals was added to the literature background as recommended by reviewer 1.

2 References were corrected to reflect changes in the body of the manuscript.

Thank you again for publishing our manuscript in the *World Journal of Diabetes*.

Sincerely yours,

Myra L. Clark, PhD, RN  
Assistant Professor of Nursing  
University of Virginia School of Nursing  
Charlottesville, Virginia, 22908 United States

Fax: +1-434-9242787

E-mail: [mlc4bf@virginia.edu](mailto:mlc4bf@virginia.edu)